```
Mauna Loa Solar Observatory Observer's Log
 ______
       Wed Jan 25 17:48:11 GMT 2006
Year: 06 Doy: 025
Observer: yasukawa
WEATHER COMMENT: Wed Jan 25 17:50:47 GMT 2006
In thick altocumulus/altostratus overcast, light S breeze, temp 40 F.
  end
**CHIP PROBLEM**: Wed Jan 25 17:51:40 GMT 2006
Replaced VIGRA GDM-9000 board that failed on 1/21 with replacement board
sent from Boulder. MCC mouse and video now working correctly.
  end
WEATHER COMMENT: Wed Jan 25 17:57:08 GMT 2006
Clear hole may open up. Opened dome to prepare for possible observing run.
  end
Wed Jan 25 18:05:16 GMT 2006
                              MKIV
                                      Start Patrol
Wed Jan 25 18:05:31 GMT 2006
                              CHIP
                                       Start Patrol
Wed Jan 25 18:05:41 GMT 2006
                              PICS
                                       Start Patrol
WEATHER COMMENT: Wed Jan 25 18:06:38 GMT 2006
Clear hole
  end
PSPT COMMENT: Wed Jan 25 18:18:04 GMT 2006
PSPT is being worked on by Bunny and Alice, no observation attempts at this
time due to predicted briefness of clear sky.
  end
A NEW PSPT EXABYTE IS LOADED ON KENOBI Wed Jan 25 18:19:12 GMT 2006
A NEW TAPE HAS BEEN PUT INTO KAIEE DLT DRIVE, Wed Jan 25 18:19:16 GMT 2006
Wed Jan 25 18:59:22 GMT 2006
                              MKIV
                                      End Patrol
Wed Jan 25 18:59:28 GMT 2006
                              MKIV
                                      Start Cal
WEATHER COMMENT: Wed Jan 25 19:04:02 GMT 2006
Hole closed up, in overcast. S wind kicked up to 13 mph.
 end
WEATHER COMMENT: Wed Jan 25 19:16:02 GMT 2006
Cap cloud formed and is starting to threaten. Closing domes.
end
Wed Jan 25 19:16:48 GMT 2006
                              CHIP
                                       End Patrol
Wed Jan 25 19:16:50 GMT 2006
                              PICS
                                       End Patrol
Wed Jan 25 19:18:11 GMT 2006
                              MKIV
                                      End Cal
Wed Jan 25 19:39:40 GMT 2006
                              MKIV
                                      End Patrol
WEATHER COMMENT: Wed Jan 25 19:40:45 GMT 2006
In light blowing mist.
end
WEATHER COMMENT: Wed Jan 25 20:30:15 GMT 2006
In heavy overcast.
end
NICE MK4
             IMAGE: 1841
```

NICE CHIP

IMAGE: 1841

```
NICE PICSLIMB IMAGE: 1842
NICE PICSDISC IMAGE: 1841
PSPT COMMENT: Wed Jan 25 23:23:22 GMT 2006
Thanks to the great work and perserverence of Alice and Bunny, we now
have PSPT booting correctly off of disk1 and once Bunny configures the
disk space to accomodate the data flow, we should be in business.
Phantomnet disconnected and realnet connected.
____end___
COMMENT: Thu Jan 26 00:55:00 GMT 2006
For Joan: 8.0 man-hours on sit today
___end_
COMMENT: Thu Jan 26 00:57:26 GMT 2006
In fog and light rain. Leaving early to get to FEDEX before they close.
____end_
Thu Jan 26 00:58:05 GMT 2006
        MkIV
18 05.rawmk4
              18 23.rawmk4
                             18 38.rawmk4
                                            18 53.rawmk4
                                                           c18 59.rawmk4
18 08.rawmk4
              18 26.rawmk4
                                            18 56.rawmk4
                                                           c19 05.rawmk4
                             18 41.rawmk4
              18_29.rawmk4
                                            19_02.rawmk4
                                                           c19_11.rawmk4
18_11.rawmk4
                             18_44.rawmk4
18_17.rawmk4
              18_32.rawmk4
                            18_47.rawmk4
                                            19_08.rawmk4
```

18 50.rawmk4

19 14.rawmk4

18 20.rawmk4

18_35.rawmk4